Author's response to reviews

Title: The influence of parents and the home environment on preschoolers' physical activity behaviours: A qualitative investigation of childcare providers’ perspectives

Authors:

Patricia Tucker (ttucker2@uwo.ca)
Melissa M van Zandvoort (melissavanzandvoort@gmail.com)
Shauna M Burke (sburke9@uwo.ca)
Jennifer D Irwin (jenirwin@uwo.ca)

Version: 5 Date: 11 March 2011

Author's response to reviews:

As per the email received on March 6, 2011 - all changes have been made to finalize the manuscript. Specifically, the abstracts have been changed to be the same, the track and change highlighting has been turned off, the country of affiliations has been added, the vertical lines in the table have been removed, the references have been updated (et al removed, and in print articles updated), and the location of the in text citation has been modified.